Comments:

1. At present, anterior cervical discectomy and fusion is the preferred treatment for cervical degenerative disc disease and is widely accepted as the gold standard. Although anterior cervical discectomy and fusion is extensively performed, it has been noted that internal fixation with a titanium plate may cause various postoperative complications, such as pseudarthrosis, loosening of the internally fixed plates, and dysphagia. In this study, the authors investigated the efficacy of anterior cervical discectomy and fusion using ROI-C for internal fixation in cervical degenerative disc disease. The study is very interesting. The methods are described in detail. Results are discussed with updated references. Some minor editing is required. And a minor language polishing should be revised.

2. This is an interesting study of the cervical degenerative disc disease treated by anterior cervical discectomy and fusion with self-locking fusion cage. The authors should include the limit of the study, and make a short discussion. Some minor misspelling words should be corrected.

Responses:
Dear Reviewer:
Thank you for your comments concerning our manuscript entitled "Outcomes of cervical degenerative disc disease treated by anterior cervical discectomy and fusion with self-locking fusion cage (ROI-C)" (Manuscript ID 73497). Those comments are all valuable and very helpful for revising and improving our paper, as well as the important guiding significance to us. We have studied comments carefully and have made the correction which we hope meet with approval.
Yours Sincerely,
Zhang Bo